**Author’s response to reviews**

**Title:** The global switch from trivalent oral polio vaccine (tOPV) to bivalent oral polio vaccine (bOPV): Facts, Experiences and Lessons Learned from the south-south zone; Nigeria, April 2016

**Authors:**

Enya Bassey Bassey (bassey69@yahoo.com; bbenya69@gmail.com)

Fiona Braka (brakaf@who.int)

Rui Gama Vaz (ruivaz@who.int)

William Komakech (komakech@who.int)

Sisay Gashu Tegegne (gashutegegns@who.int)

Sylvester Toritseju Maleghemi (maleghemis@who.int)

Richard I Koko (kokor@who.int)

Thompson U Igbu (igbut@who.int)

Faith Ireye (ireyef@who.int)

Sylvester O Agwai (agwais@who.int)

Godwin Ubong Akpan (akpang@who.int)

Abdul-Aziz Garba Mohammed (mohammedabd@who.int)

Angela Okocha-Ejeko (peaches_spears@rocketmail.com)

**Version:** 5  **Date:** 05 Jan 2018

**Author’s response to reviews:**

Cecilia Devoto, BioMed Central (BMC) Infectious Diseases. Dear Sir, In response to the reviewer’s comment on the manuscript titled “The global switch from trivalent Oral Polio Vaccine (tOPV) to bivalent Oral Polio Vaccine (bOPV): Facts, experiences and Lessons Learned from the south-south zone; Nigeria, April 2016”, the authors have made the recommended changes outlined in the table below:

<table>
<thead>
<tr>
<th>Reviewer 1</th>
<th>S/N</th>
<th>Comments</th>
<th>Authors Response</th>
</tr>
</thead>
<tbody>
<tr>
<td>1</td>
<td></td>
<td>Please make sure that the manuscript text is correctly formatted as per our submission guidelines; in particular, please rename &quot;Introduction&quot; as &quot;Background&quot;, and include a &quot;Conclusions&quot; heading after the Discussion. See manuscript</td>
<td>2</td>
</tr>
</tbody>
</table>

Please make sure that the Declarations section is complete and includes all the sub-headings and in the same order as detailed below. If any item is not relevant for your manuscript, please write "Not
applicable" underneath the sub-heading.

Please re-phrase your Availability of data and materials section. As only summarized data is presented in this manuscript, we ask that you amend the data availability statement. This statement refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please either include all raw data obtained in this study or clearly state that the data is available upon request and who should be contacted if someone wants to request the data.

Please include a statement at the end of the Authors' contributions section to confirm that all authors have read and approved the final version of the manuscript.

Figures: please remove figures from the main text and provide figures as separate, individual files. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them. Figures have been removed from manuscript and presented as individual files.

Additional files: we note that the information included here is identical to "Figure 1" and the additional files can therefore be removed, unless you prefer to present this as an additional file, in which case Figure 1 should be removed and the figures re-numbered accordingly. If you wish to keep additional files, please include a List of additional files for all additional files you would like published, as per guidelines here: https://bmcpregnancychildbirth.biomedcentral.com/submission-guidelines/preparing-your-manuscript#preparing+additional+files

The additional files can be removed.

Please state in the cover letter whether the map depicted in Figure 1 is your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. If the above conditions are not met the image needs to be removed. Please note the editors may request proof of permission at any time. The authors own the map. It was made using basic Microsoft word tools.

Thank you very much for your kind consideration and I look forward to hearing back from the journal soonest.

Warm Regards,
Dr Bassey Enya Bassey